Author’s response to reviews

Title: A review of clinical trial designs used to detect a disease-modifying effect of drug therapy in Alzheimer's disease and Parkinson's disease

Authors:

David McGhee (mailto:davidjmmcghee@gmail.com; mailto:david.mcghee@abdn.ac.uk)

Craig Ritchie (mailto:craig.ritchie@ed.ac.uk)

John Zajicek (jz45@st-andrew.ac.uk)

Carl Counsell (carl.counsell@abdn.ac.uk)

Version: 2 Date: 13 Mar 2016

Author’s response to reviews:

Editorial Requests:

(1) Ethics. If your study involves humans, human data or animals, then your article should contain an ethics statement which includes the name of the committee that approved your study. If ethics was not required for your study, then this should be clear stated and a rationale provided.

• We have inserted the following sentence in the beginning of the methodology section: ‘As this study was a systematic review ethical approval was not required.’

(2) Consent. If your article is a prospective study involving human participants then your article should include a statement detailing consent for participation. If individual clinical data is presented in your article, then you must clarify whether consent for publication of these data was obtained.

• Our article is a systematic review and, therefore no consent from individuals was required or obtained.

• All data within our article is already published elsewhere and fully reference within our manuscript.

(3) Availability of supporting data. BioMed Central strongly encourages all data sets on which the conclusions of the paper rely be either deposited in publicly available repositories (where available and appropriate) or presented in the main papers or additional supporting files, in machine-readable format whenever possible. Authors must include an Availability of Data and Materials section in their article detailing where the data supporting their findings can
be found. The Accession Numbers of any nucleic acid sequences, protein sequences or atomic coordinates cited in the manuscript must be provided and include the corresponding database name.

- We have included an ‘Availability of data and materials’ section in the article to make this clear. It states: ‘The datasets supporting the conclusions of this article are included within the article and its additional files.’

(4) Authors Contributions. Your 'Authors Contributions' section must detail the individual contribution for each individual author listed on your manuscript.

- We have ensured that this is made clear in the Authors Contributions’ section.